Author’s response to reviews

Title: Why are eligible patients not prescribed aspirin in primary care? A qualitative study indicating measures for improvement.

Authors:

Dr Duncan Short (d.short@keele.ac.uk)
Martin Frischer (m.frischer@keele.ac.uk)
James NR Bashford (j.n.r.bashford@keele.ac.uk)
Darren Ashcroft (darren.ashcroft@man.ac.uk)

Version: 3 Date: 10 Jul 2003

The following changes suggested in the email have been made:
"- Country - Please remember to include country in the affiliations' details.
- Qualifications - Please remove the authors' qualifications from the title page
- References - Though they are in the correct format, they should not be superscript to text."

The reviewers did not suggest any other amendments - both indicated "accept without revision"